Author’s response to reviews

Title: The ThRombolysis in UnconTrolled Hypertension (TRUTH) protocol: an observational study on treatment strategy of elevated blood pressure in stroke patients eligible for IVT

Authors:
Thomas Zonneveld (t.p.zonneveld@amc.nl)
Ale Algra (a.algra@umcutrecht.nl)
Diederik Dippel (d.dippel@erasmusms.nl)
Jaap Kapelle (l.kappelle@umcutrecht.nl)
Robert van Oostenbrugge (r.van.oostenbrugge@mumc.nl)
Yvo Roos (y.b.roos@amc.nl)
Marieke Wermer (m.j.h.wermer@lumc.nl)
Bart van der Worp (h.b.vanderworp@umcutrecht.nl)
Paul Nederkoorn (p.j.nederkoorn@amc.nl)
Nyika Kruyt (ndkruyt@gmail.com)

Version: 1 Date: 06 Oct 2015

Author’s response to reviews:

Dear Editor, dear Ying Lou,

Thank you very much for your response and suggested corrections.

We have updated our cover letter, which now states that the TRUTH study is ongoing, externally funded by the NutsOhra Foundation, and was reviewed by the Medical Ethical Review Committee of the Academic Medical Center in Amsterdam and by local ethical or scientific committees in all participating centers. We have sent documents with these statements to BMCSeriesEditorial@biomedcentral.com.

If any further information is needed on these subjects, please contact us again.

Best regards,
Nyika Kruyt